Author's response to reviews

Title: Medical students are afraid to include abortion in their future practices: in-depth interviews in Maharastra, India.

Authors:

   Susanne Sjöström (susanne.sjostrom@ki.se)
   Birgitta Essén (birgitta.essen@kbh.uu.se)
   Kristina Gemzell (kristina.gemzell@ki.se)
   Marie Klingberg-Allvin (mkl@du.se)

Version: 6 Date: 1 October 2015

Author's response to reviews: see over
Dear Editor-in-Chief,

Thank you for accepting our manuscript Medical students are afraid to provide abortion- in depth interviews in Maharastra, India, manuscript number MS: 1932624389171709. Approval to conduct the study and disseminate the result was obtained in writing from the following colleges:

1. Lokmanya Tilak Medical College and Hospital, Sion, Mumbai. Dr Chaturvedi Nov 20 2012
2. Dr. D.Y. Patil Medical College, Pimpri, Pune, Dr Bhawalkar, Nov 22 2012
3. Indira Gandhi Medical College Nagpur, Dr Kubde Feb 13 2013
4. R.C.S.M Government Medical College Kolapur, Dr Gandham, April 17 2013
5. Government Medical College, Miraj, Dr Mrs Naik, April 18 2013
6. Rural Medical College Loni, Ahmednagar, Dr Phalke April 19 2013

To ensure the anonymity of participants we have chosen to omit the institutional names in the manuscript.

Sincerely yours,

Susanne Sjöström, MD
Corresponding author
susanne.sjostrom@ki.se

email addresses of authors:

Birgitta Essén birgitta.essen@kbh.uu.se
Kristina Gemzell Danielsson Kristina.Gemzell@ki.se
Marie Klingberg Allvin mkl@du.se